Author’s response to reviews

Title: Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Authors:
Ming-wei Liu (lmw2004210@163.com)
Rui Wei (758437603@qq.com)
Mei-xian Su (Kmsmx24@126.com)
Hui Li (3420220794@qq.com)
Tian-wen Fang (13890801975@163.com)
Wei Zhang (2436730979@qq.com)

Version: 4 Date: 11 Jan 2018

Author’s response to reviews:

The Biomed Central Editorial Team 09 January 2018

BCAM-D-17-00608R3 Effects of Panax notoginseng saponins on severe acute pancreatitis through the regulation of mTOR/Akt and caspase-3 signaling pathway by upregulating miR-181b expression in rats

Ming-wei Liu; Rui Wei; Mei-xian Su; Hui Li; Tian-wen Fang; Wei Zhang

BMC Complementary and Alternative Medicine

Thank you for consideration of our manuscript for publication in your journal.

We have reviewed the above manuscript according to your reviewer’s comments.

Responses to the comments of Reviewer

1. Due to the list of authors for your manuscript changing, we would appreciate it if you could please submit a change of authorship form with your revision.
We have submitted a change of authorship form with our revision.

2. Please clearly explain in your Methods section how the animals were killed for your experiment.

The animals were euthanized by cervical dislocation in our Methods section.